Author’s response to reviews

Title: Bubble CPAP to support preterm infants in rural Rwanda: a retrospective cohort study

Authors:

Evrard Nahimana (evnahimana@gmail.com)
Masudi Ngendahayo (ngemasudi@gmail.com)
Hema Magge (hema.magge@gmail.com)
Jackline Odhiambo (ojacklineatieno@gmail.com)
Ernest Muhirwa (ernestinoy@gmail.com)
Jean Nepo Uwilingiyemungu (ujineposs@yahoo.fr)
Fulgence Nkikabahizi (fuldgence@gmail.com)
Regis Habimana (habiregis@hotmail.fr)
Cheryl L. Amoroso (cheryl.amoroso@gmail.com)
Bethany L Hedt-Gauthier (bethhedt@gmail.com)

Version: 5
Date: 29 May 2015

Author’s response to reviews:

May 29th, 2015

To the Editors of BMC Pediatrics:

We are pleased to submit the updated manuscript entitled “Bubble CPAP to support preterm infants in rural Rwanda: a retrospective cohort study”. Per your additional feedback to the second submission, we made the following changes:

1. Heading for table 2 was added
2. Legend for figure 1 was added

Thank you for your kind consideration and we look forward to hear your feedback

Sincerely,

Evrard Nahimana
District Clinical Director, Kirehe
Partners In Health-Inshuti Mu Buzima